\begin{table}%t1
\caption{\label{Tabdata}Galaxies with HI data.}
\begin{tabular}{cccccccccccccc}
\hline\hline
\noalign{\smallskip}
Name         &  Type                   &$a$	& $b$    &  Mag	  &	 Vel   &  Dist  &Memb&    SHI	& WHIc  & MHI	& Ref.$^a$&HI--def&  Wqual\\
	     &			       &'	&'     & 	  & km s$^{-1}$&  Mpc	&    &Jy~km~s$^{-1}$&km s$^{-1}$& $M_{\odot}$&&     &       \\
\hline
NGC~3712      &  SB?                    &1.70	& 0.60 &  14.90   &	 1580  &   -- 	& Bg &    7.60  & 119	& 9.04  &   4	 &--0.07&    1\\
UGC~6599      &  Ibm                    &1.20	& 0.70 &  16.50   &	 1569  &   -- 	& Bg &    5.31  & 104	& 8.88  &   2	 &--0.12&    5\\
UGC~6610      &  Scd:                   &2.10	& 0.40 &  15.00   &	 1851  &   -- 	& Bg &    12.76 & 199	& 9.39  &   1	 &--0.20&    1\\
UGC~6637      &  S0                     &0.90	& 0.46 &  15.00   &	 1836  &   -- 	& Bg &    1.84  & 132	& 8.54  &   15   &--0.11&    5\\
UGC~6684      &  IA?(s)mV:              &1.00	& 0.40 &  15.70   &	 1788  &   -- 	& Bg &    5.78  & 124	& 9.02  &   2	 &--0.30&    1\\
IC2957       &  S0-a                   &0.60	& 0.40 &  14.80   &	 1776  &  8.6   & Bg &    2.24  & 88	& 8.60  &   1	 &--0.51&    3\\
UGC~6782      &  ImV                    &2.00	& 2.00 &  15.00   &	 525   &   -- 	& CI &    6.96  & 83	& 8.54  &   1	 & 0.21&    1\\
NGC~3900      &  SA(r)0+                &3.16	& 1.70 &  12.50   &	 1798  &   -- 	& Bg &    16.84 & 420	& 9.48  &   1	 &--0.10&    1\\
UGC~6791      &  Scd:                   &2.05	& 1.41 &  14.99   &	 1852  &  29.3  & Bg &    5.10  & 212	& 8.99  &   1	 & 0.18&    1\\
NGC~3899      &  SAB(s)b?pec            &2.45	& 1.07 &  13.21   &	 1779  &  10.4  & Bg &    4.87  & 159	& 8.94  &   2	 & 0.24&    2\\
UGC~6881      &  Im:                    &1.40	& 0.70 &  17.00   &	 607   &   -- 	& CI &    2.42  & 66	& 8.08  &   2	 & 0.45&    2\\
UGC~6900      &  Sd                     &2.00	& 1.29 &  14.80   &	 590   &   -- 	& CI &    1.69  & 83	& 7.92  &   2	 & 0.82&    1\\
NGC~4020      &  Sbd?sp                 &2.00	& 0.73 &  13.28   &	 760   &  14.2  & CI &    14.64 & 161	& 8.84  &   1	 &--0.11&    1\\
CGCG157080   &  Irr                    &0.58	& 0.42 &  15.00   &	 714   &   -- 	& CI &    2.64  & 125	& 8.12  &   3	 &--0.12&    2\\
NGC~4032      &  Im:                    &1.90	& 1.80 &  12.81   &	 1268  &   -- 	& CI &    22.10 &  94	& 9.04  &   5	 &--0.33&    2\\
UGC~7007      &  SB(s)mIV               &0.84	& 0.79 &  15.50   &	 774   &   -- 	& CI &    2.90  &  36	& 8.16  &   2	 & 0.06&    2\\
NGC~4062      &  SA(s)c                 &1.59	& 1.16 &  12.50   &	 769   &  25.5  & CI &    24.13 & 293	& 9.57  &   1	 &--0.54&    1\\
NGC~4080      &  Im?                    &1.20	& 0.50 &  14.28   &	 567   &  10.9  & CI &    2.86  & 138	& 7.90  &   2	 & 0.36&    1\\
IC2992       &  S0?                    &0.60	& 0.60 &  14.80   &	 580   &   -- 	& CI &    1.42  & --	& 7.85  &   15   &--0.25&    5\\
UGC~7131      &  Sdm:                   &1.50	& 0.40 &  15.10   &	 253   &   -- 	& CI &    4.56  & 100	& 8.35  &   2	 & 0.22&    1\\
NGC~4136      &  SAB(r)c                &4.00	& 3.70 &  11.69   &	 609   &   -- 	& CI &    39.51 & 84	& 9.29  &   6	 & 0.00&    1\\
NGC~4173      &  Sdm:                   &5.04	& 0.69 &  13.59   &	 1127  &   -- 	& CI &    42.12 & 75	& 9.32  &   7	 &--0.01&    1\\
UGC~7236      &  Im:                    &1.10	& 0.90 &  16.28   &	 945   &   -- 	& CI &    4.75  & 57	& 8.37  &   8	 & 0.01&    4\\
NGC~4203      &  SAB0-                  &3.40	& 3.20 &  11.80   &	 1086  &   -- 	& CI &    48.11 & 234	& 9.38  &   9	 &--0.44&    1\\
NGC~4204      &  SB(s)dm                &3.60	& 2.90 &  12.90   &	 856   &   -- 	& CI &    34.28 & 65	& 9.23  &   7	 &--0.13&    1\\
MRK1315      &  BCD                    &1.05	& 0.60 &  16.50   &	 847   &   -- 	& CI &    20.89 & -	& 9.02  &   15   &--0.66&    5\\
UGC~7300      &  Im                     &1.40	& 1.20 &  14.90   &	 1210  &   -- 	& CI &    14.51 & 77	& 8.86  &   1	 &--0.33&    1\\
UGC~7321      &  Sd                     &6.74	& 0.43 &  14.00   &	 408   &  21.7  & CI &    44.51 & 217	& 9.69  &   1	 & 0.04&    1\\
NGC~4245      &  SB(r)0/a:              &2.90	& 2.20 &  12.31   &	 815   &  15.3  & CI &    0.76  & 214	& 7.62  &   1	 & 1.23&    1\\
NGC~4274      &  (R)SB(r)ab             &6.80	& 2.50 &  11.34   &	 930   &  17.1  & CI &    9.91  & 432	& 8.83  &   10   & 0.72&    1\\
NGC~4278      &  E1-2                   &4.10	& 3.80 &  11.20   &	 649   &12.6$^b$& CI &    10.52 & 395	& 8.59  &   16   & 0.38&    2\\
NGC~4286      &  SA(r)0/a:              &1.60	& 1.00 &  14.10   &	 644   &   -- 	& CI &    0.71  & --	& 7.55  &   15   & 0.81&    5\\
UGC~7428      &  Im:                    &1.30	& 1.20 &  14.10   &	 1137  &   -- 	& CI &    7.35  & 49	& 8.56  &   2	 &--0.08&    2\\
IC3215       &  Sdm:                   &1.26	& 0.35 &  15.00   &	 1019  &   -- 	& CI &    2.64  & 102	& 8.12  &   2	 & 0.35&    2\\
NGC~4310      &  (R')SAB(r)0+?          &2.20	& 1.20 &  13.22   &	 913   &  10.4  & CI &    1.15  & 166	& 7.47  &   11   & 0.88&    2\\
NGC~4314	     &	SB(rs)a 	       &4.20	& 3.70 &  11.43   &	 963   &   -- 	& CI &    0.40  & 111	& 7.30  &   1	 & 1.92&    1\\
IC3247	     &	Sd      	       &1.86	& 0.26 &  15.25   &	 569   &  23.0  & CI &    3.50  & 152	& 8.64  &   1	 & 0.34&    1\\
NGC~4359	     &	SB(rs)c?sp             &3.50	& 0.80 &  13.40   &	 1253  &  -- 	& CI &    14.72 & --	& 8.86  &   15   & 0.33&    0\\
IC3308	     &	Sdm:    	       &1.30	& 0.20 &  15.41   &	 316   &   -- 	& CI &    13.30 & 131	& 8.82  &   3	 &--0.34&    2\\
NGC~4395	     &	SA(s)m: 	       &13.20	& 11.00&  10.64   &	 319   &  13.9  & CI &    401.61& 117	& 10.26 &   1	 &--0.40&    1\\
NGC~4393	     &	SABd    	       &3.20	& 3.00 &  12.70   &	 755   &   -- 	& CI &    25.35 & 92	& 9.10  &   15   &--0.07&    5\\
NGC~4414	     &	SA(rs)c?	       &3.60	& 2.00 &  10.96   &	 716   &19.1$^b$& CI &    70.20 & 347	& 9.78  &   1	 &--0.36&    1\\
UGC~7584	     &	Sdm:    	       &0.90	& 0.50 &  16.00   &	 603   &   -- 	& CI &    3.65  & 33	& 8.26  &   1	 & 0.00&    1\\
NGC~4448	     &	SB(r)ab 	       &3.90	& 1.40 &  12.00   &	 661   &  18.4  & CI &    3.00  & 377	& 8.38  &   1	 & 0.93&    1\\
NGC~4455	     &	SB(s)d?sp	       &2.80	& 0.80 &  12.93   &	 637   &   -- 	& CI &    30.67 & 98	& 9.18  &   1	 &--0.23&    1\\
UGC~7673	     &	ImIII-IV	       &1.40	& 1.30 &  15.28   &	 642   &   -- 	& CI &    10.38 & 73	& 8.71  &   1	 &--0.18&    1\\
UGC~7698	     &	Im      	       &6.50	& 4.50 &  13.00   &	 331   &   -- 	& CI &    40.98 & 59	& 9.31  &   1	 & 0.15&    1\\
NGC~4509	     &	Sab pec?      	       &0.90	& 0.60 &  14.10   &	 937   &  6.1   & CI &    6.49  & 57	& 7.81  &   9	 & 0.21&    2\\
NGC~4525	     &	Scd:    	       &2.37	& 0.96 &  13.40   &	 1172  &  9.2   & CI &    9.94  & 141	& 8.29  &   1	 & 0.28&    1\\
NGC~4562	     &	SB(s)dm:sp             &2.32	& 0.69 &  13.90   &	 1353  &   -- 	& CI &    8.29  & 121	& 8.61  &   1	 & 0.22&    1\\
NGC~4559	     &	SAB(rs)cd	       &10.70	& 4.40 &  10.46   &	 816   &  7.2   & CI &    340.70& 236	& 9.62  &   1	 &--0.28&    1\\
IC3571	     &	BCD     	       &0.60	& 0.50 &  17.79   &	 1260  &  -- 	& CI &    0.91  & --& 7.65  &   17   & 0.36&    5\\
NGC~4565	     &	SA(s)b?sp	       &15.90	& 1.85 &  10.42   &	 1230  &13.4$^b$& CI &    341.11& 394	& 10.16 &   1	 &--0.12&    3\\
UGC~A292	     &	ImIV-V  	       &1.00	& 0.70 &  16.00   &	 308   &   -- 	& CI &    15.64 & 28	& 8.89  &   1	 &--0.57&    1\\
NGC~4631	     &	SB(s)d  	       &15.50	& 2.70 &   9.75   &	 606   &  5.3   & CI &   1077.60& 276	& 9.86  &   12   &--0.48&    1\\
MCG+06-28-022&	BCD		       &0.68	& 0.58 &  15.40   &	 892   &   -- 	& CI &    1.42  &--	& 7.85  &   15   & 0.24&    5\\
NGC~4656	     &	SB(s)m pec    	       &6.46	& 0.66 &  10.60   &	 664   &   -- 	& CI &    190.55& --	& 9.98  &   15   &--0.52&    5\\
UGC~A294	     &	S0 pec	    	       &0.70	& 0.40 &  15.10   &	 947   &   -- 	& CI &    6.89  & 84	& 8.53  &   3	 &--0.82&    2\\
NGC~4670	     &	SB(s)0/a pec:BCD       &1.40	& 1.10 &  13.09   &	 1069  &  20.9  & CI &   15.35  & 121	& 8.88  &   13   &--0.35&    1\\
UGC~A298	     &	BCD?    	       &0.50	& 0.30 &  15.50   &	 801   &  1.9   & CI &    0.73  & 38	& 7.56  &   14   & 0.34&    2\\
NGC~4725	     &	SAB(r)ab pec	       &10.70	& 7.60 &  10.11   &	 1206  &13.0$^b$& CI &    141.19& 382	& 9.75  &   1	 &--0.10&    1\\
UGC~7990	     &	Im	  	       &1.00	& 0.60 &  17.00   &	 512   &   -- 	& CI &    2.38  & 70	& 8.07  &   2	 & 0.25&    1\\
KUG1249+263  &	Irr		       &0.60	& 0.40 &  16.72   &	 1225  &   -- 	& CI &    2.33  & --	& 8.06  &   15   &--0.05&    5\\
NGC~4747	     &	Sbcd? pec sp  	       &3.50	& 1.20 &  12.96   &	 1190  &  10.7  & CI &    39.00 & 155	& 9.02  &   8	 &--0.12&    3\\
IC3840	     &	Sm/Irr: 	       &1.00	& 0.24 &  16.90   &	 583   &   --	& CI &    2.23  & --	& 8.04  &   15   & 0.28&    5\\
UGC~8011	     &	Im      	       &1.40	& 1.10 &  18.00   &	 776   &   -- 	& CI &    10.18 & 116	& 8.70  &   1	 &--0.17&    1\\
MRK1338	     &	E	    	       &0.40	& 0.20 &  15.50   &	 1069  &  -- 	& CI &    0.68  & 19	& 7.53  &   14   &--0.24&    2\\
NGC~4789A     &	IB(s)mIV-V             &3.00	& 2.20 &  13.94   &	 374   &  19.7  & CI &    31.77 & 70	& 9.46  &   1	 &--0.28&    1\\
UGC~8030	     &	Im:     	       &0.83	& 0.44 &  18.00   &	 628   &   -- 	& CI &    1.36  & 34	& 7.83  &   15   & 0.38&    2\\
NGC~4826	     &	(R)SA(rs)ab            &10.00	& 5.40 &   9.36   &	 408   &  4.9   & CI &    51.25 & 308	& 8.47  &   1	 & 0.65&    1\\
UGC~8181	     &	Sdm:    	       &1.50	& 0.40 &  16.00   &	 886   &   -- 	& CI &    3.99  & 80	& 8.30  &   2	 & 0.27&    1\\
UGC~8333	     &	Im:     	       &1.30	& 0.40 &  17.00   &	 935   &   -- 	& CI &    11.88 & 125	& 8.77  &   1	 &--0.29&    1\\						\hline
\end{tabular}
\smallskip
Notes: $a$: references to HI data:
1~=~Springob et~al. \cite{Springob05};
2~=~Schneider et~al. (\cite{Schneider90});
3~=~Scodeggio \& Gavazzi (\cite{Scodeggio93});
4~=~Theureau et~al. (\cite{Theureau98});
5~=~Helou et~al. (\cite{Helou84});
6~=~Lewis (\cite{Lewis87});
7~=~Fisher \& Tully (\cite{Fisher81});
8~=~Huchtmeier \& Richter (\cite{Huchtmeier89});
9~=~Noordermeer et~al. (\cite{Noordermeer05});
10~=~Huchtmeier (\cite{Huchtmeier82});
11~=~Chamaraux et~al. (\cite{Chamaraux87});
12~=~Rots (\cite{Rots80});
13~=~Thuan (\cite{Thuan81});
%14~=~Eder et~al. (1991);
14 = Huchtmeier et~al. (\cite{Huchtmeier05});
15 = Hyperleda;
%16~=~de Vaucouleurs et~al. (1991);
16~=~Burstein et~al. (\cite{Burstein87});
17~=~Dahlem et~al. (\cite{Dahlem05})\\
$b$: from Ferrarese et~al. (\cite{Ferrarese00}).
\end{table}